Author's response to reviews

Title: Strengthening health workforce capacity through modular work-based training

Authors:

Joseph KB Matovu Mr (jmatovu@musphcdc.ac.ug)
Rhoda K Wanyenze Dr (rwanyenze@musphcdc.ac.ug)
Susan Mawemuko Ms (susanne@musphcdc.ac.ug)
Olico Okui Dr (ookui@musph.ac.ug)
William Bazeyo Dr (wbazeyo@musph.ac.ug)
David Serwadda Dr (dserwada@imul.com)

Version: 2 Date: 24 July 2012

Author's response to reviews: see over
July 23, 2012

Dr Emily Crow
Executive Editor
BMC International Health & Human Rights

Dear Editor,

Re: Manuscript Re-submission

Please receive our revised Manuscript entitled, “Strengthening health workforce capacity through modular work-based training” for possible publication in your prestigious Journal, BMC International Health and Human Rights.

We have responded to the initial comments as follows:

a) Turning the Manuscript into a ‘Correspondence Article’
   We have changed the article into a ‘Correspondence Article’ and have inserted the words ‘Correspondence Article’ at the beginning of the main text. See page 3.

b) Ethical issues pertaining to the quotations cited
   We have inserted a statement indicating that the program evaluation exercise from which the quotations indicated were cited received ethical approval from Makerere University School of Public Health ethical committee. See page 8.

We expect that the revised Manuscript will be well received by your readers, and look forward to receiving additional guidance on the way forward with regard to its possible publication in BMC International Health and Human Rights Journal.

Yours sincerely,

Joseph KB Matovu
Corresponding author